Author's response to reviews

Title: Quantitative assessment of the asphericity of pretherapeutic FDG uptake as an independent predictor of outcome in NSCLC

Authors:

Ivayla Apostolova (ivayla.apostolova@med.ovgu.de)
Julian Rogasch (julianrogasch@gmx.de)
Ralph Buchert (ralph.buchert@charite.de)
Heinz Wertzel (h.wertzel@lungenklinik-lostau.de)
H. Jost Achenbach (jost.achenbach@lungenklinik-lostau.de)
Jens Schreiber (jens.schreiber@med.ovgu.de)
Sandra Riedel (sandra.riedel@med.ovgu.de)
Christian Furth (christian.furth@med.ovgu.de)
Alexandr Lougovski (A.Lougovski@hzdr.de)
Georg Schramm (G.Schramm@hzdr.de)
Frank Hofheinz (F.Hofheinz@hzdr.de)
Holger Amthauer (holger.amthauer@med.ovgu.de)
Ingo G Steffen (ingo.steffen@med.ovgu.de)

Version: 5 Date: 15 November 2014

Author’s response to reviews:

EDITORIAL REQUEST:

1. “Please include the name of the institutional review board which approved this study in the methods section of your manuscript.”

Reply / Changes in the manuscript:

The study protocol had been approved by the Ethics Committee of the University Hospital Magdeburg A. ö. R. at the Otto-von-Guericke University (reference number, 159/13; RAD233) and complied with the Declaration of Helsinki. Changes were made in section Patients in Methods of the revised manuscript.